Author’s response to reviews

Title: Subspecialty surveillance of long-term course of small and moderate muscular ventricular septal defect: heterogenous practices, low yield.

Authors:

Erik L Frandsen (ketch87@hotmail.com)
Aswathy V House (asha.vaikomhouse@unmc.edu)
Yunbin Xiao (xiaoyunbinshirui@gmail.com)
David A Danford (ddanford@childrensomaha.org)
Shelby Kutty (skutty@unmc.edu)

Version: 4 Date: 24 September 2014

Author’s response to reviews:

1. TITLE PAGE:
   Please include the email addresses of all authors on the title page. Please ensure that it is identical to the one entered on the submission system. Also, kindly remove the suffixes from the title page.
   Response: Done (Page 1)

2. ETHICS:
   Please update your ethics statement to include the name of the ethics committee that approved your study.
   Response: Done (Page 5, paragraph 2)